Author’s response to reviews

Title: Fulminant encephalopathy in a child with hyperferritinemic sepsis: a case report

Authors:
Lilin Huang (921295597@qq.com)
Shumei Peng (pengsm2010@163.com)
Ronghan Li (150348200@qq.com)
Danyu Xie (13580362856@163.com)
Dongping Huang (hdpaaa@126.com)

Version: 4 Date: 20 Feb 2020

Author’s response to reviews:

Point-by-point responses to editor:

1. Please amend the consent for publication statement so that it reads: "Written informed consent was obtained from the parents of the patient for publication of this case report"

Response: We are really sorry for the mistake. The consent for publication statement has been amended in line 260-261, page 9 in Consent for publication which is described as below: “Written informed consent was obtained from the parents of the patient for publication of this case report.”

2. Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).

Currently, the contributions of author SP does not automatically qualify them for authorship. In the section “Authors’ contributions”, please provide further clarifications on their contributions, and see our guidelines for authorship below.

An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. Authors are expected to fulfil the criteria below (adapted from McNutt et al., Proceedings of the National Academy of Sciences, Feb 2018, 201715374; DOI: 10.1073/pnas.1715374115; licensed under CC BY 4.0): Each author is expected to have made substantial contributions to the conception OR design of the work; OR the acquisition, analysis, OR interpretation of data; OR the creation of new software used in the work; OR have drafted the work or substantively revised it AND to have approved the submitted version (and any substantially modified version that involves the author's contribution to the study); AND to have agreed both to be personally accountable for the author's own contributions and to ensure that questions related to the accuracy or integrity of any part of the work, even ones in which the author was not personally involved, are appropriately investigated, resolved, and the resolution documented in
the literature.
Acquisition of funding, collection of data or general supervision of the research group, alone, does not usually justify authorship.

Response: We are sorry for the mistake. The contributions of authors have been amended in line 276-279, page 10 in Authors’ contributions which is described as below:
“Study concept and design: Lilin Huang. Acquisition of data: Ronghan Li and Danyu Xie. Analysis and interpretation of data: Lilin Huang and Dongping Huang. Drafting of the manuscript: Lilin Huang, Shumei Peng and Dongping Huang. All authors have read and approved the manuscript.”

3. Please remove the editing certificate from the file inventory, as it is no longer needed.

Response: Thank you very much for your professional suggestion. The editing certificate from the file inventory has been removed.

4. Please proofread and ensure that when you upload your revised submission it is in the final form for publication. Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only.

Response: Thanks for your professional suggestion. The tracked changes or highlighting in the revised manuscript have been removed and a single clean copy of the manuscript has been uploaded in the final form for publication.